Author's response to reviews

Title: Health and life insurance as an alternative to malpractice tort law

Authors:

Walton Sumner II (wsumner@dom.wustl.edu)

Version: 3 Date: 24 December 2009

Author's response to reviews:

The reference formatting was flawed for two blocks of citations. These are corrected in the new attachment.